% Math 370 hw7 Spring 2014
%
\documentclass{article}
%\usepackage[textheight=10in, textwidth=7.5in]{geometry}
\usepackage[textheight=10in]{geometry}
\pagestyle{empty}

\usepackage{amsmath}
\usepackage{amsthm}
\usepackage{hyperref}
\usepackage{graphicx}
\usepackage{verbatim}

%% create an environment for theorems
\newtheorem*{thm}{Theorem}

\newcommand{\coursehome}
{http://www.cs.umb.edu/~eb/370}

\title{History of Mathematics \\
Homework 8
}
\author{Ethan Bolker}

\begin{document}

\maketitle


\begin{enumerate}

\item Term paper due dates

\begin{itemize}

\item Thursday, May 1: paper draft due in class and \LaTeX{} in dropbox.
I will get comments back to you over the weekend.

\item Thursday, May 8: paper draft due in class to trade with classmates
  for feedback. Presentation draft due in class for me to comment
  on. (Slides aren't required, but if you want them you may want to
  learn to use Beamer
  (\url{http://en.wikipedia.org/wiki/Beamer\_\%28LaTeX\%29}). 

\item Tuesday, May 13: last class. Feedback to your classmates due.

\item Tuesday, May 20, 11:30-2:30. Presentations, in room W01-0031
  (our regular classroom).

\item Exercise 3.25 in our text.

\item Build an elegant timeline for the history of mathematics we've
covered this semester. (I will discuss this in class.)

\end{itemize}


\end{enumerate}

\end{document}
